Author’s response to reviews

Title: Availability of Essential Medicines and Pharmaceutical Inventory Management Practice at health centers of Adama town, Ethiopia

Authors:

Adane Kefale (adanet2011@gmail.com)

Hafiza Shebo (hafizahayredin@yahoo.com)

Version: 4 Date: 04 Apr 2019

Author’s response to reviews:

Authors’ response for technical comments

We thank for your constructive comments. We have made the following revisions

Abstract: minor topographic revision (line 33)

Background:

Paragraph 1: the definition of EMs rephrased

Paragraph 3: the paragraph is modified and statements rephrased

Paragraph 5, line 72-74, minor revision on the text

Declaration: modified as per the comments

Figure legends: moved after references according to the comment